Author’s response to reviews

Title: Establishing a genetic link between FTO and VDR gene polymorphisms and obesity in the Emirati population

Authors:

Saad Khan (SMK452@student.bham.ac.uk)
Sara Chehadeh (sarah.chehadeh@kustar.ac.ae)
Mehera Abdulrahman (maRad@dha.gov.ae)
Wael Osman (wael.osman@kustar.ac.ae)
Habiba ALsafar (habiba.alsafar@kustar.ac.ae)

Version: 2 Date: 24 Nov 2017

Author’s response to reviews:

Manuscript Number: MGTC-D-17-00048

RE: Submission of Revised Manuscript - Establishing a genetic link between FTO and VDR gene polymorphisms and obesity in the Emirati population.

Dear Dr. Matteo Pasini,

We are grateful for reviewing our original manuscript for the BMC Medical Genetics journal. We would also like to thank Dr. Francesca Marchegiani for her time, valuable comments and constructive suggestions to help improve the quality of our manuscript.

We have made all the necessary changes to address all the technical comments.

Thank you for all your help and support
Regards

Habiba Alsafar